

Citation: Gibson-Helm ME, Bailie J, Matthews V, Laycock AF, Boyle JA, Bailie RS (2018) Identifying evidence-practice gaps and strategies for improvement in Aboriginal and Torres Strait Islander maternal health care. PLoS ONE 13(2): e0192262. https://doi.org/10.1371/journal. pone.0192262

Editor: Ganesh Dangal, National Academy of Medical Sciences, NEPAL

Received: July 27, 2017

Accepted: January 11, 2018

Published: February 7, 2018

Copyright: © 2018 Gibson-Helm et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Research involving Aboriginal and Torres Strait Islander peoples is strictly governed in Australia by the Australian Code for the Responsible Conduct of Research and Values and Ethics: Guidelines for Ethical Conduct in Aboriginal and Torres Strait Islander Health Research. In order to comply with these national regulations, both the legal agreements governing health service participation in the ABCD National Research Partnership and the human research ethics committee approvals of multiple RESEARCH ARTICLE

Identifying evidence-practice gaps and strategies for improvement in Aboriginal and Torres Strait Islander maternal health care

Melanie E. Gibson-Helm¹*, Jodie Bailie², Veronica Matthews³, Alison F. Laycock³, Jacqueline A. Boyle¹, Ross S. Bailie²

1 Monash Centre for Health Research and Implementation, School of Public Health and Preventive Medicine, Monash University, Melbourne, Victoria, Australia, 2 University Centre for Rural Health, The University of Sydney, Lismore, New South Wales, Australia, 3 Menzies School of Health Research, Charles Darwin University, Brisbane, Queensland, Australia

* melanie.gibson@monash.edu

Abstract

Introduction

Adverse pregnancy outcomes are more common among Aboriginal and Torres Strait Islander populations than non-Indigenous populations in Australia. Later in life, most of the difference in life expectancy between Aboriginal and Torres Strait Islander women and non-Indigenous women is due to non-communicable diseases (NCDs). Most Aboriginal and Torres Strait Islander women attend health services regularly during pregnancy. Providing high-quality care within these appointments has an important role to play in improving the current and future health of women and babies.

Aim

This study engaged stakeholders in a theory-informed process to use aggregated continuous quality improvement (CQI) data to identify 1) priority evidence-practice gaps in Aboriginal and Torres Strait Islander maternal health care, 2) barriers and enablers to high-quality care, and 3) strategies to address identified priorities.

Methods

Three phases of reporting and feedback were implemented using de-identified CQI data from 91 health services between 2007 and 2014 (4,402 client records). Stakeholders (n = 172) from a range of professions and organisations participated.

Results

Stakeholders identified four priority areas relating to NCDs: smoking, alcohol, psychosocial wellbeing and nutrition. Barriers or enablers to high-quality care included workforce support, professional development, teamwork, woman-centred care, decision support, equipment and community engagement. Strategies to address the priorities included upskilling staff to provide best practice care in priority areas, advocating for availability of healthy food,



jurisdictions preclude the sharing of a de-identified data set without evidence that it's proposed use addresses the ethical requirements of Reciprocity, Respect, Equality, Responsibility, Survival and Protection, and Spirit and integrity. Data requests addressing these criteria may be sent to the ABCD National Research Partnership Data Access Committee care of Professor Ross Bailie (ross. bailie@sydney.edu.au).

Funding: The National Health and Medical Research Council (www.nhmrc.gov.au) funded the ABCD National Research Partnership Project (#545267) and the Centre for Research Excellence in Integrated Quality Improvement (#1078927). Inkind and financial support has been provided by the Lowitja Institute (www.lowitja.org.au) and a range of Community-Controlled and Government agencies. M Gibson-Helm and J Boyle are National Health and Medical Research Council Fellows (#1110701 and #1122540). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

housing and local referral options, partnering with communities on health promotion projects, systems to facilitate continuity of care and clear referral pathways.

Conclusions

This novel use of large-scale aggregate CQI data facilitated stakeholder input on priority evidence-practice gaps in maternal health care in Australia. Evidence-practice gaps relating to NCD risk factors and social determinants of health were prioritised, and stakeholders suggested both healthcare-focussed initiatives and approaches involving the community and the wider health sector. The findings can inform health service planning, advocacy, interagency strategies, and future research.

Introduction

Approximately 4% of all women giving birth in Australia are Aboriginal and/or Torres Strait Islander (12,817 in 2014) [1]. While Aboriginal and Torres Strait Islander women tend to have babies at younger ages than non-Indigenous women [1], risk factors for non-communicable diseases (NCDs) are already common among young Aboriginal and Torres Strait Islander women [2]. Smoking, high-risk alcohol use, obesity, underweight, poor nutrition and gestational diabetes are risk factors for both adverse pregnancy outcomes and NCDs that are more common among Aboriginal and Torres Strait Islander women [1, 3–7]. This is consistent with a life expectancy of 10 years less for Aboriginal and Torres Strait Islander women, predominantly due to NCDs [5].

While many Aboriginal and Torres Strait Islander women have healthy pregnancies and babies, considerable disparities in perinatal outcomes, such as low birth weight and preterm birth, persist compared to the general population in Australia [1]. These outcomes are associated with neonatal morbidity and mortality, and may increase the risk of the baby developing NCDs in later life [8, 9]. High levels of psychological distress, and stressors during pregnancy, are common among Aboriginal and Torres Strait Islander women and may also affect the next generation through associations with low birth weight and preterm birth [10–12].

NCD risk factor prevalence in Aboriginal and Torres Strait Islander populations is associated with socio-economic disadvantage and the intergenerational effects of colonisation [5–7]. Social determinants of health and wellbeing also affect access to health care and the availability of opportunities to optimise health [6]. Aboriginal and Torres Strait Islander women are more likely than non-Indigenous women to live in remote locations and in locations of the lowest socioeconomic status [1]. However, despite these and other substantial challenges [13, 14], 86% of Aboriginal and Torres Strait Islander women attend five or more pregnancy care visits [1]. These are important opportunities to reduce disparities and improve both current and future health by providing all components of recommended maternal health care. While many aspects of maternal health care are provided at high levels by primary health centres serving Aboriginal and Torres Strait Islander communities, there is a widespread need to ensure all recommended services are delivered [15–17].

One strategy used to drive evidence-based care is Continuous Quality Improvement (CQI), a management approach that aims to constantly increase the efficiency and effectiveness of organizational systems and processes [18, 19]. Maternal health care in Australia is provided in settings that differ in characteristics such as population density, health centre governance structures, jurisdictional policies, and availability of local health and social services, and often

involves a range of health professionals. For example, women in remote locations usually attend primary health care (PHC) centres for pregnancy care and then transfer to regional hospitals for the birth. Governance of Aboriginal and Torres Strait Islander specific PHC centres is either through the state/territory government health department or a community board [20].

Data collected by individual PHC centres as part of their CQI activities can be aggregated and used to identify trends across health services and over time [21], including aspects of care that are delivered at low levels by many health centres and across different settings. Such patterns are likely to represent inadequacies in the broader healthcare delivery system, and require strategies for change at multiple levels of the system rather than solely actions by individual health centres [22, 23]. Thus, the analysis of aggregated data can contribute to health service planning, policy development and future research programs if disseminated in a way that has meaning to the intended audience. However, timely and relevant translation of research evidence into policy and clinical practice is a common challenge in the health sector [24–27].

We hypothesise that one way to generate research evidence that is better matched to the needs of health services, policy makers and affected populations [25], and more likely to result in multidisciplinary strategies leading to large-scale improvement [23], is to incorporate the perspectives of multiple stakeholders when interpreting analyses of aggregated CQI data. Therefore, we aimed to facilitate stakeholder-identification of 1) priority evidence-practice gaps in Aboriginal and Torres Strait Islander maternal health care, 2) barriers and enablers to addressing such gaps and 3) strategies for achieving improvement. A secondary aim was to map the suggested strategies back to both the barriers/enablers and to the evidence-practice gaps.

Materials and methods

Study design

The "Engaging Stakeholders in Identifying Priority Evidence-Practice Gaps and Strategies for Improvement in PHC" (ESP) Project was designed to combine evidence on how to achieve large-scale change with knowledge co-creation [22]. This was an implementation research study design employing a theory-informed, iterative and interactive dissemination method. The theoretical basis and methods for the overall ESP Project have been reported in detail previously [22]. Feedback and experience from earlier iterations of the ESP Project (child health, chronic illness and preventive care) were used to inform the maternal health ESP process.

Setting and context of the ESP Project: ABCD National Research Partnership

The ESP Project drew on aggregated CQI data collected as part of the Audit and Best Practice for Chronic Disease (ABCD) National Research Partnership–a collaboration that brought together PHC services, policy, support organisations and research institutions to guide and support collaborative CQI research to improve the quality of Aboriginal and Torres Strait Islander PHC across Australia. The ABCD Partnership linked more than 170 PHC centres in different parts of Australia in collaborative CQI research. The protocol and outcomes have been reported in detail previously [15–17, 21, 28] and the tools and protocols are available online [29]. Briefly, the maternal health audit tool was developed by an expert working group (e.g. medical officers, nurses and educators) and included consultation of health centre staff who would be using the tools. The audit tool was based on best practice guidelines (e.g. the Women's Business Manual [30] and the Three Centres Consensus Guidelines on Antenatal Care [31], policy and research reports [32–36]. Between 2007 and 2014, 91 Aboriginal and Torres Strait Islander PHC centres provided clinical audit and systems assessment data on the delivery of maternal health care to the ABCD Partnership project. Participating PHC centres were located in five of Australia's eight mainland states or territories and, reflecting the diverse settings of maternal health care in Australia, a range of locations, governance structures and population sizes were represented (S1 Table). To be eligible for inclusion in the audit, a client must: have an infant between 2 and 14 months; have been resident in the community for 6 months of the infant's gestation; and be expected to use the health service as her usual source of PHC. The audit protocol included sampling guidelines to generate a sample likely to reflect the general population of clients [15, 28]. The evidence-based systems assessment tool guided a facilitated consensus process enabling health centre teams to evaluate the state of development of the PHC centre's organisational systems and processes [28, 37].

ESP Project participant recruitment: All phases

Each phase of the ESP Project used the same recruitment methods, aiming to engage stakeholders in different roles across the health care system. Stakeholders were identified by drawing on the extensive networks of the ABCD National Research Partnership, the Centre for Research Excellence in Integrated Quality Improvement and the ESP Project research team. These networks included health practitioners (e.g. doctors, midwives, Aboriginal health workers, nurses), managers and policy makers at various levels of the health system, researchers, and staff of peak bodies and support organisations that represent the interests of communitycontrolled health services and Aboriginal and Torres Strait Islander communities. Stakeholders were recruited by email invitation to individuals and organisations with known interest in Aboriginal and Torres Strait Islander maternal health care (e.g. community-controlled health organisations, universities and the Australian Institute of Health and Welfare), professional society newsletters (e.g. the Primary Health Care Research & Information ebulletin), and websites (e.g. Menzies School of Health Research and the Australian Indigenous HealthInfoNet). A snowballing distribution technique was used, whereby all stakeholders initially contacted were requested to forward the ESP reports and links to the online questionnaires through their networks to increase ESP Project reach. To enable engagement of those less likely to provide individual responses, we encouraged responses from facilitated group discussions.

Ethics

Ethical approval for the ABCD National Research Partnership was obtained from research ethics committees in each relevant Australian jurisdiction [22]. All participants in the ESP Project provided individual informed consent.

Variables, data sources and measurement

In all phases the online questionnaires could be completed by individuals or on behalf of a group of people. A group facilitation guide was provided to assist with group discussion. The first section of each questionnaire covered information about the stakeholder including their profession, organisation and jurisdiction. The specific data collection tools and methods for each phase are described in the following paragraphs.

Phase one: Identify and prioritise current evidence-practice gaps. In Phase One stakeholders were asked to identify priority evidence-practice gaps for maternal health care (Phase One outcome) using cross-sectional, national CQI data. The data comprised the most recent maternal health audit completed by health centres during the period 2012–2014 (n = 65 health centres; 1091 audit records; and 58 systems assessments). Prior to stakeholder recruitment, the research team used this data to develop a preliminary set of priorities based on the following criteria:

- 1. Aspects of maternal health care that were generally recorded at low levels.
- 2. Aspects of care where there was wide variation between health centres in recorded delivery of care.
- 3. Aspects of care that were being delivered at high levels by most health centres, but that were being delivered at much lower levels by some health centres.
- 4. Organisational systems that were relatively less developed.

The data and preliminary priorities were then circulated as a report and a plain language summary with a link to an online questionnaire (S1 File). The questionnaire asked stakeholders to rate the relative importance of the 33 preliminary priority evidence-practice gaps. Stakeholders were asked to use a scale of 1–10, with 10 being the most important, however they were also able to give equal ratings to multiple priorities.

Phase Two: Identify barriers, enablers and strategies for addressing the prioritised evidence-practice gaps. In Phase Two stakeholders were asked to identify barriers and enablers and strategies for best practice maternal health care (Phase Two outcomes). In this phase, stakeholders were provided with longitudinal, national CQI data for the evidence-practice gaps prioritised in Phase One (from 2007 to 2014: 4402 audited records and 242 systems assessments from 91 PHC centres), and a brief on the evidence from previous quality improvement research, including research in Aboriginal and Torres Strait Islander PHC. A plain language summary of the full report, jurisdiction-specific data supplements and a link to the Phase Two questionnaire (S2 File) were also provided.

The questionnaire asked stakeholders to consider the longitudinal data, the evidence brief and their experience in PHC, and then to rate the quality of staff and health centre or system attributes relevant to supporting best practice in maternal health care, particularly relating to the priority evidence-practice gaps. A five-point Likert scale was used: strongly disagree that the attribute is in place, partly disagree, partly agree, strongly agree, I don't know. The list of attributes that could be potential enablers (if in place) or barriers (if not in place) to improvement was drawn from international and national research [38–41] and has been published previously [38].

The questionnaire then asked stakeholders to suggest "strategies and actions that could be used to address barriers and enablers for each priority evidence-practice gap" using free-text comment boxes.

Phase Three: Final report preparation and feedback. A final draft report was prepared and sent to stakeholders, summarising the CQI data for the prioritised evidence-practice gaps, the barriers and enablers stakeholders agreed were important for addressing the priority evidence-practice gaps, and the strategies that stakeholders suggested. An online questionnaire was also provided, asking for any additional comments or suggestions on the presented barriers, enablers and strategies for maternal health care (Phase Three outcome) (S3 File). This provided an opportunity for stakeholders to confirm that the final report accurately reflected their responses to the Phase One and Two questionnaires.

Data analysis

Analysis of questionnaire responses. Questionnaire responses were analysed after each phase so that the data could inform the next phase. Individual and group responses were analysed together. In Phase One the aspects of maternal health care that received importance

ratings of 8–10 by more than 90% of stakeholders were considered the prioritised evidencepractice gaps. In Phase Two if two-thirds of stakeholders strongly or partly agreed that an attribute was in place it was considered a current enabler. If two-thirds of stakeholders strongly or partly disagreed that an attribute was in place it was considered a current barrier. For the suggested strategies (free text) one of the ESP Project team (MGH) collated and summarised the strategies and actions. A second ESP Project researcher (JB) then checked the summary against the raw questionnaire data to ensure accuracy. In Phase Three any new barriers, enablers or strategies suggested by stakeholders were incorporated into the final report by one ESP Project team member (MGH) and checked by a second (JB).

Synthesis of findings. One final piece of synthesis was conducted once all phases had been completed. The ESP project had guided stakeholders through a process, starting with prioritising evidence-practice gaps, then identifying barriers and enablers, and finally suggesting strategies. At each stage stakeholders were encouraged to consider the findings and their responses from the previous step so that the stages built on each other. Therefore, we hypothesised that the suggested strategies could be mapped back to both the barriers/enablers and to the evidence-practice gaps and attempted to organise the findings to map these relationships. The synthesis drew on the multidisciplinary expertise of the ESP Project team: quality improvement, Aboriginal and Torres Strait Islander health, obstetrics and gynaecology, primary health care, public health, management, knowledge translation and implementation. One of the ESP Project team members (MGH) conducted the initial mapping by hypothesising links between the suggested strategies and specific barriers/enablers and evidence-practice gaps. The ESP Project team reviewed and refined the mapping to generate the final version. This mapping allowed us to assess whether stakeholders had indeed kept the barriers and enablers in mind when discussing strategies to address the evidence-practice gaps.

Results

Participants

Fifty-one discrete responses, representing over one hundred stakeholders, were received across the three online questionnaires (41 individual and 10 group responses). The represented organisations included community-controlled (n = 19), government (n = 12) and general practice health services (n = 3), research organisations (n = 13), government health departments (n = 13), and health service support organisations (n = 9). Participants included nurses and midwives (n = 30), doctors (n = 18), Aboriginal and/or Torres Strait Islander Health Practitioners or Workers (n = 10), managers and board members (n = 10), CQI facilitators (n = 3), researchers (n = 10) and others (n = 8). Five individual respondents identified as Aboriginal and/or Torres Strait Islander, and all groups had some Aboriginal and/or Torres Strait Islander members. Numbers may not tally with the total number of responses, as respondents were able to select multiple answers.

Phase One

Approximately 112 people (27 individuals and 6 group responses on behalf of 85 people) participated in Phase One.

The aggregated CQI data showed that many key aspects of maternal health care were provided at high levels by most health centres. Examples included laboratory investigations related to maternal immunity and infection, follow-up for abnormal glucose tests and anaemia, measurement of blood pressure, fundal height and foetal heart rate, and discussion of breastfeeding and contraception at the postnatal visit. Stakeholders identified eight evidence-practice gaps as priorities for improvement (Figs 1 and 2). These priorities related to smoking and alcohol, psychosocial wellbeing, Sudden Unexpected Death in Infancy (SUDI) and nutrition. A few stakeholders also highlighted quality of care indicators that were not captured by the audit data: family wellbeing and family support during pregnancy and at the time of birth, and pre-conception health care.

Phase Two

Approximately 60 stakeholders (10 individuals and 3 group responses on behalf of approximately 50 people) participated in Phase Two. The attributes of health centres, the health system and staff that the majority of stakeholders agreed or disagreed were in place are presented in Table 1.

Enablers of best practice maternal health care, particularly regarding the priority evidencepractice gaps, were systems that support adequate health centre equipment, staff training and development, teamwork, decision support resources and systems for health centre staffing and workforce support for some staff types (Table 1). Systems requiring strengthening in order to better support best practice maternal health care included systems to build and support the Aboriginal and/or Torres Strait Islander workforce, woman-centred care, and community capacity and engagement (Table 1). The attributes of staff in place to support best practice maternal health care, as relevant to the priority evidence-practice gaps, reflected the domains of knowledge, skills, professional identity, beliefs about capabilities, optimism, beliefs about consequences, intentions, memory, attention and decision support, and social influence (Table 1).

Stakeholders used the free-text sections to provide context for their ratings of health centre, system and staff attributes, and to suggest strategies to address barriers and enablers for each prioritised evidence-practice gap. The suggested strategies are summarised in Fig 3 (centre column, labelled A-J). Some stakeholders specifically noted the importance of understanding the social determinants of health of each woman and in her local area: housing availability and conditions, food availability and accessibility, transport, and safety at home. A related theme was the availability of effective, practical and sustainable risk factor interventions. The importance of midwives in conducting and coordinating maternal health care for both well women and women with complex care needs was also emphasised. A need specifically noted was for adequate systems to support the wellbeing of the workforce, particularly around workloads, time, backfill and reflective practice.

Some strategies addressed health centre and staff attributes such as training, continuity of care, and processes to support consistent care. Other strategies built on the existing strong staff attributes but require partnerships across the broader health system such as enhancing referral processes, advocating for local services and working with communities on health promotion and education projects. An overarching theme across all strategies was the importance of working with communities and in a culturally appropriate manner.

Phase Three

Five responses were received regarding review of the draft final report (4 individuals and 1 group response on behalf of 5 people), providing feedback from nine people in a variety of roles. An additional strategy suggested at this stage was to take a long-term funding approach to midwifery-led maternal health care and Aboriginal and Torres Strait Islander women's health care to enable successful temporary services or programs to continue. A second suggested strategy was to strengthen systems to recruit, retain and support doctors and nurses, especially Aboriginal or Torres Strait Islander doctors and nurses, with language skills and cultural knowledge. All respondents indicated that the ESP process had improved their understanding of how to use aggregated CQI data to inform decision-making.





https://doi.org/10.1371/journal.pone.0192262.g001

Synthesis of findings

As hypothesised, the strategies (centre column of Fig 3) could be mapped to attributes of health centres, the health system and individuals that stakeholders had identified as current barriers or enablers (left column of Fig 3) and also mapped to the evidence-practice gaps (right column

- 4. Screening all pregnant women for emotional wellbeing:
 - (91% regarded this a priority area)
- 5. Following-up with women identified as 'at risk' based on emotional wellbeing assessment in pregnancy: (94% regarded this a priority area)



- 6. At the postnatal visit, discussing the importance of keeping a safe environment for the baby, especially when sleeping: (94% regarded this a priority area)
- 7. At the postnatal visit, discussing smoking and how a smoking environment increases the risk of Sudden Infant Death Syndrome (SIDS): (91% regarded this a priority area)



Fig 2. Prioritised evidence-practice gaps in Aboriginal and Torres Strait Islander maternal health care and provision of these services 2007–2014, continued: Assessment of emotional wellbeing, discussion of Sudden Unexpected Death in Infancy risk reduction, and discussion of nutrition. n = number of health centres; patient records.

https://doi.org/10.1371/journal.pone.0192262.g002

Table 1. Attributes of health centres, the health system and staff that are currently enablers or barriers to best practice maternal health care for Aboriginal and Torres Strait Islander women, particularly across the prioritised evidence-practice gaps^a.

Health centre and health system attributes	Agreed already in place	Attribute domain
Current enablers (attributes that are already in place)		
Systems to support staff development.	9/13 respondents ^b	Training and development
PHC staff function effectively in teams.	9/13	Teamwork
PHC staff are clear about their roles in relation to other team members.	9/13	
PHC centres generally have appropriate numbers of:		
• medical specialists	9/13	Staffing/workforce support, recruitment and retention
administrative staff	10/13	
Systems to recruit, retain and support general practitioners.	9/13	
PHC centres generally have adequate equipment.	8/12	Equipment, finance, resources and facilities
Availability of best practice guidelines and other decision support resources.	8/11	Decision support
Staff are adequately trained to use these resources.	8/11	
Current barriers (attributes that require strengthening)		
Systems to:		Staffing/workforce support, recruitment and retention
• Ensure appropriate numbers of Aboriginal and/or Torres Strait Islander Health Practitioners /Workers and midwives.	2/13	
• Recruit, retain and support Aboriginal and/or Torres Strait Islander Health Practitioners /Workers.	3/13	
• Ensure staff have support from experienced staff, especially when health centres are affected by staff turnover and shortages.	2/12	
Systems that support all PHC staff to:		Patient-centred care
• Understand the needs and aspirations of people living in Aboriginal and Torres Strait Islander communities for the purpose of providing best practice maternal health care.	3/11	
• Provide care that is respectful of and responsive to individual patient preferences, needs and values and for patient values to guide all clinical decisions.	2/11	
Systems to:		Community capacity, engagement and mobilisation
• Increase the expectation of community members regarding best practice care.	2/10	
• Enhance the health literacy of community members.	1/10	
• Build the capability of PHC staff and to support them to develop effective links and to work in partnership with communities.	2/10	
Current staff attributes	Agreed already in place	Attribute domain
Enablers	·	
Staff are confident in their ability to provide best practice care.	8/9	Beliefs about capabilities
Staff know the content and objective of best practice care.	7/9	
Staff know how to provide best practice care.	6/9	Knowledge
Staff recognise the professional responsibility to provide best practice maternal health care for Aboriginal and Torres Strait Islander women.	7/9	Professional identity
Staff believe that best practice maternal health care will have benefits at a population level.	7/9	Beliefs about consequences

(Continued)

Table 1. (Continued)

Health centre and health system attributes	Agreed already in place	Attribute domain
Staff often or always remember to provide best practice care.	7/9	Memory, attention and decision processes
Staff have no trouble focussing their attention on providing best practice care.	7/9	
Staff believe that most people of influence in health centres are seen to support provision of best practice maternal health care to Aboriginal and Torres Strait Islander women.	7/9	Social influences
Staff have the skills to provide best practice care.	6/9	Skills
Staff are optimistic about the future with regard to providing best practice maternal health care for Aboriginal and Torres Strait Islander women.	6/9	Optimism
Staff have a very strong intention to provide best practice care every day.	6/9	Intentions

^aSmoking and alcohol, psychosocial wellbeing, Sudden Unexpected Death in Infancy and nutrition. ^bActual number of stakeholder views reflected here is higher (up to approximately 60 people) than the presented

denominator as each group response is only included once.

https://doi.org/10.1371/journal.pone.0192262.t001

of Fig 3). All of the strategies could be mapped to more than one priority area and to both individual and higher level attributes. For example, advocating for local services to refer women to (strategy I) could build on the belief that best practice care will have benefits at the population level (beliefs about consequences) and could also increase the expectation of community members regarding best practice care (community capacity, engagement and mobilization). Ultimately, this strategy could address several prioritised evidence-practice gaps by improving provision of follow-up for multiple social determinants of health (high-risk alcohol use, social and emotional wellbeing).

Discussion

Summary of findings

This research engaged a range of stakeholders across the health sector to identify eight priority evidence-practice gaps in Aboriginal and Torres Strait Islander maternal health care. These gaps related to smoking, alcohol, nutrition, and social and emotional wellbeing. Current enablers of best practice care included decision support resources and systems that support adequate equipment, training and development, and teamwork. Many staff attributes were also identified as current enablers. Systems perceived to be less developed (and therefore current barriers) included those that support the Aboriginal and/or Torres Strait Islander workforce, woman-centred care, and community engagement. Strategies to address the evidence-practice gaps related to training, referral processes, continuity of care, processes to support consistent and comprehensive care, advocacy for availability of local services and working with communities on health promotion and education projects.

Interpretation and comparison with existing literature

Aboriginal and Torres Strait Islander women are populations at increased risk of NCDs and many young women will enter pregnancy with existing NCDs or NCD risk factors. Six of the eight prioritised evidence-practice gaps directly related to NCD risk factors and social determinants of health. Therefore, we conclude that stakeholders in this research recognised these elevated risks and the important role of maternal health care in reducing NCD risk for both



Fig 3. Strategies to address the barriers and enablers for each prioritised evidence-practice gap. Centre column: Strategies or actions suggested by stakeholders during Phase Two (labelled A-J). Left column: attributes of health centres, the health system and staff identified as barriers or enablers during Phase Two. Letters A-J indicate which strategies map to these attributes. Right column: Evidence-practice gaps prioritised during Phase One. Letters A-J indicate which strategies map to these priority areas.

https://doi.org/10.1371/journal.pone.0192262.g003

mothers and babies. Addressing social determinants of health requires both improving health care at individual health centres and broader approaches. Therefore, it's not surprising that some strategies addressed attributes of individuals and health centres or the health system. However, such strategies require partnerships with communities, external services and policy makers. This fits with the hypothesis that evidence-practice gaps across many health centres are due to failures of the wider health system [22] and require a combined program of strategies simultaneously targeting different parts of the health system [23]. Recent diabetes care research identified different patterns of delivery and improvement for different types of care processes (e.g. laboratory tests, patient education) [42], supporting the need for action at different levels of the heath system.

As shown in Fig.3, stakeholders often identified the same or similar strategies to address different evidence-practice gaps, such as training and upskilling staff to have conversations with women about smoking, alcohol use and emotional wellbeing (strategy E). This is consistent with the focus of CQI on the functioning of organisational systems and identifying root causes rather than on isolated problems, individual components of care or individual staff [18, 19]. The barriers and enablers identified here, such as systems and processes that support woman centred-care, staff training and development, and multidisciplinary teamwork, are consistent with a review identifying key organisational elements that facilitate high-quality performance in PHC [43]. One notable difference is the additional emphasis that stakeholders in our research placed on strengthening systems that support community capacity, engagement and mobilisation. This may reflect the setting of the ESP Project (PHC for Aboriginal and Torres Strait Islander populations) but is likely to be relevant to PHC for other culturally and linguistically diverse populations. Some of the barriers and enablers identified in our research were the same as those identified in other areas of Aboriginal and Torres Strait Islander PHC [38], yet there were also some differences, perhaps reflecting the views of a different sample of stakeholders or the qualitative differences between maternal health care and care for disease states.

There is a growing emphasis on the importance of meaningful stakeholder involvement in setting research priorities and co-creation of implementation research projects, both generally and specifically in PHC [44–48]. The process described here adds to this developing field by using large-scale data regarding health care delivery as a starting point for priority setting. The finding that each strategy could be matched to barriers/enablers and to evidence-practice gaps indicates that stakeholders drew on both when proposing strategies. This suggests the process was consistent with the implementation research approach of using evidence (the CQI data) and theory (individual and higher level attributes) to develop strategies for change [49]. However, further formal evaluation of the ESP program is needed to confirm this. Best practice in research priority setting and healthcare improvement also includes consultation of patients [44]. This aligns with our finding of a need to strengthen systems that support community engagement.

Implications for future research and clinical practice

Dissemination of research evidence and timely implementation in policy and clinical practice is a key challenge for researchers, policy makers, health services and health professionals [24–26]. Our findings can be used for health service planning. By matching service performance data to common barriers and enablers and then to possible strategies, the new insights generated here can contribute to operational planning at health centres. At a strategic planning level, the findings are a call to action for health services to commit to a maternal health strategy that recognises the critical role of comprehensive maternal health care in improving pregnancy outcomes in the short term and reducing NCD risk for both mothers and babies in the long term.

The findings can also be a starting point for strategies to be developed collaboratively across different levels of the health system. For many of the prioritised gaps a key strategy was to improve the availability of local, appropriate services and resources to support health services to take a life-course and woman-centred approach to maternal health care. The findings relating to current service performance data and stakeholder-identified barriers and enablers address a challenge to evidence uptake frequently reported by policy makers: inadequate access to timely, relevant and high-quality research evidence [27].

The ESP Project process created an important opportunity for stakeholders to contribute their perspectives to research. Capturing these perspectives enables researchers to focus on areas important to people involved in health service delivery. The process and findings may facilitate researchers and stakeholders working together to develop research programs with real world application.

Strengths and limitations

As barriers exist across multiple levels of the health system, we included questions on broader health centre and system attributes in the Phase Two questionnaire. These additional questions have not been validated like the questions relating to individual attributes. Another limitation is that recruitment relied on stakeholders passing on the invitation through their networks, so it is not possible to accurately measure reach or response rates. Caution is needed when interpreting the findings from the Phase Two questionnaire as relatively fewer responses were received, which limits generalisability.

One of the main strengths of this research is that it used the largest and most comprehensive longitudinal dataset relating to Aboriginal and Torres Strait Islander maternal health care provision. However, it is important to note that the dataset captured recorded delivery and may underestimate actual service delivery due to under-documentation. Another strength is that a range of stakeholders was recruited and so the findings reflect knowledge of front-line health workers and others working within the health system.

Conclusion

This novel study used large-scale aggregated CQI data and a theory-informed dissemination strategy to engage stakeholders involved in Aboriginal and Torres Strait Islander maternal health care in Australia. Stakeholders identified priority evidence-practice gaps in maternal health care relating to NCD risk factors and social determinants of health, some barriers and enablers associated with these gaps, and strategies for achieving improvement. These findings can be used in health service planning, advocacy efforts, development of inter-agency strategies, and to inform the maternal health care research agenda.

Supporting information

S1 File. Phase One survey to identify and prioritise current evidence-practice gaps in Aboriginal and Torres Strait Islander maternal health care. (PDF)

S2 File. Phase Two survey to identify barriers, enablers and strategies for addressing the prioritised evidence-practice gaps. (PDF)

S3 File. Phase Three survey asking stakeholders for confirmation that the final report accurately reflected their responses to the previous surveys. (PDF) **S1 Table.** Characteristics of PHC centres providing maternal health audit data from 2007 to 2014. PHC: primary health care, CQI: continuous quality improvement, n: number. (DOCX)

Acknowledgments

The development of this manuscript would not have been possible without the active support, enthusiasm and commitment of staff in participating primary health care services, and members of the ABCD National Research Partnership and the Centre for Research Excellence in Integrated Quality Improvement. We thank everyone who participated in the ESP Maternal Health Project. We also thank Perri Hull for report preparation and dissemination.

Author Contributions

Conceptualization: Ross S. Bailie.

Data curation: Jodie Bailie, Veronica Matthews.

Formal analysis: Melanie E. Gibson-Helm, Veronica Matthews, Jacqueline A. Boyle.

Funding acquisition: Ross S. Bailie.

Investigation: Melanie E. Gibson-Helm, Jodie Bailie, Jacqueline A. Boyle.

Methodology: Jodie Bailie, Alison F. Laycock, Ross S. Bailie.

Project administration: Jodie Bailie.

Resources: Ross S. Bailie.

Supervision: Jacqueline A. Boyle, Ross S. Bailie.

Visualization: Veronica Matthews.

Writing - original draft: Melanie E. Gibson-Helm.

Writing – review & editing: Jodie Bailie, Veronica Matthews, Alison F. Laycock, Jacqueline A. Boyle, Ross S. Bailie.

References

- 1. Australian Institute of Health and Welfare. Australia's mothers and babies 2014—in brief. Canberra: AIHW, 2016.
- Crinall B, Boyle J, Gibson-Helm M, Esler D, Larkins S, Bailie R. Cardiovascular disease risk in young Indigenous Australians: a snapshot of current preventive health care. Australian and New Zealand Journal of Public Health. 2016:n/a-n/a. https://doi.org/10.1111/1753-6405.12547 PMID: 27372907
- Abell SK, Nankervis A, Khan KS, Teede HJ. Type 1 and Type 2 Diabetes Preconception and in Pregnancy: Health Impacts, Influence of Obesity and Lifestyle, and Principles of Management. Semin Reprod Med. 2016; 34(02):110–20. Epub 08.02.2016. https://doi.org/10.1055/s-0035-1571196 PMID: 26854707
- Lee AJ, Hiscock RJ, Wein P, Walker SP, Permezel M. Gestational diabetes mellitus: Clinical predictors and long-term risk of developing type 2 Diabetes—A retrospective cohort study using survival analysis. Diabetes Care. 2007; 30(4):878–83. Epub 2007/03/30. https://doi.org/10.2337/dc06-1816 PMID: 17392549.
- 5. Australian Institute of Health and Welfare. Contribution of chronic disease to the gap in adult mortality between Aboriginal and Torres Strait Islander and other Australians. Canberra, Australia: Australian Institute of Health and Welfare, 2010.
- 6. Australian Institute of Health and Welfare. The health and welfare of Australia's Aboriginal and Torres Strait Islander people Canberra, Australia: Australian Institute of Health and Welfare, 2011.

- 7. Australian Indigenous HealthInfoNet. Overview of Aboriginal and Torres Strait Islander health status 2015. Perth, Western Australia: Australian Indigenous HealthInfoNet, 2016.
- 8. Saigal S, Doyle LW. An overview of mortality and sequelae of preterm birth from infancy to adulthood. Lancet. 2008; 371(9608):261–9. http://dx.doi.org/10.1016/S0140-6736(08)60136-1. PMID: 18207020
- McMillen IC, Robinson JS. Developmental origins of the metabolic syndrome: Prediction, plasticity, and programming. Physiological Reviews. 2005; 85(2):571–633. https://doi.org/10.1152/physrev.00053. 2003 PMID: 15788706
- Prandl KJ, Rooney R, Bishop BJ. Mental health of Australian Aboriginal women during pregnancy: identifying the gaps. Archives of Women's Mental Health. 2012; 15(3):149–54. Epub 2012/04/06. doi: 10. 1007/s00737-012-0276-0. PMID: 22476364
- Cunningham J, Paradies YC. Socio-demographic factors and psychological distress in Indigenous and non-Indigenous Australian adults aged 18–64 years: Analysis of national survey data. BMC Public Health. 2012; 12(1). https://doi.org/10.1186/1471-2458-12-95 PMID: 22296820
- 12. Ding XX, Wu YL, Xu SJ, Zhu RP, Jia XM, Zhang SF, et al. Maternal anxiety during pregnancy and adverse birth outcomes: A systematic review and meta-analysis of prospective cohort studies. Journal of Affective Disorders. 2014; 159:103–10. https://doi.org/10.1016/j.jad.2014.02.027 PMID: 24679397
- **13.** Australian Health Ministers' Advisory Council. Clinical practice guidelines: antenatal care-module 1. Canberra, Australia: Australian Government Department of Health and Ageing, 2012.
- 14. Reibel T, Walker R. Antenatal services for Aboriginal women: The relevance of cultural competence. Quality in Primary Care. 2010; 18(1):65–74. Epub 2010/04/03. PMID: 20359414.
- Rumbold AR, Bailie RS, Si D, Dowden MC, Kennedy CM, Cox RJ, et al. Delivery of maternal health care in Indigenous primary care services: baseline data for an ongoing quality improvement initiative. BMC Pregnancy Childbirth. 2011; 11:16. Epub 2011/03/10. https://doi.org/10.1186/1471-2393-11-16 PMID: 21385387; PubMed Central PMCID: PMC3066246.
- Gibson-Helm ME, Rumbold AR, Teede HJ, Ranasinha S, Bailie RS, Boyle JA. Improving the provision of pregnancy care for Aboriginal and Torres Strait Islander women: a continuous quality improvement initiative. BMC Pregnancy Childbirth. 2016; 16:118. Epub 2016/05/26. https://doi.org/10.1186/s12884-016-0892-1 PMID: 27221473; PubMed Central PMCID: PMC4878071.
- Gibson-Helm ME, Teede HJ, Rumbold AR, Ranasinha S, Bailie RS, Boyle JA. Continuous quality improvement and metabolic screening during pregnancy at primary health centres attended by Aboriginal and Torres Strait Islander women. Med J Aust. 2015; 203(9):369–70. Epub 2015/10/30. PMID: 26510809.
- Laffel G, Blumenthal D. The case for using industrial quality management science in health care organizations. Journal of the American Medical Association. 1989; 262(20):2869–73. PMID: 2810623
- Berwick DM. Continuous improvement as an ideal in the health care. New Engl J Med. 1989; 320 (1):53–6. https://doi.org/10.1056/NEJM198901053200110 PMID: 2909878
- National Aboriginal Community Controlled Health Organisation. NACCHO Aboriginal Health Definitions 2014 [cited 2015 11/12/2015]. Available from: http://www.naccho.org.au/aboriginal-health/definitions/.
- Bailie R, Matthews V, Brands J, Schierhout G. A systems-based partnership learning model for strengthening primary healthcare. Implement Sci. 2013; 8(1):143. Epub 2013/12/19. <u>https://doi.org/10. 1186/1748-5908-8-143 PMID: 24344640; PubMed Central PMCID: PMC3878728.</u>
- Laycock A, Bailie J, Matthews V, Bailie R. Interactive Dissemination: Engaging Stakeholders in the Use of Aggregated Quality Improvement Data for System-Wide Change in Australian Indigenous Primary Health Care. Frontiers in public health. 2016; 4:84. Epub 2016/05/21. https://doi.org/10.3389/fpubh. 2016.00084 PMID: 27200337; PubMed Central PMCID: PMC4853415.
- 23. Ferlie EB, Shortell SM. Improving the Quality of Health Care in the United Kingdom and the United States: A Framework for Change. Milbank Q. 2001; 79(2):281–315. https://doi.org/10.1111/1468-0009. 00206 PMID: 11439467
- 24. Morris ZS, Wooding S, Grant J. The answer is 17 years, what is the question: Understanding time lags in translational research. J R Soc Med. 2011; 104(12):510–20. https://doi.org/10.1258/jrsm.2011. 110180 PMID: 22179294
- Chalmers I, Glasziou P. Avoidable waste in the production and reporting of research evidence. Lancet. 2009; 374(9683):86–9. https://doi.org/10.1016/S0140-6736(09)60329-9 PMID: 19525005
- 26. McGlynn EA, Asch SM, Adams J, Keesey J, Hicks J, DeCristofaro A, et al. The quality of health care delivered to adults in the United States. New Engl J Med. 2003; 348(26):2635–45. <u>https://doi.org/10.1056/NEJMsa022615</u> PMID: 12826639
- Oliver K, Innvar S, Lorenc T, Woodman J, Thomas J. A systematic review of barriers to and facilitators of the use of evidence by policymakers. BMC Health Serv Res. 2014; 14. https://doi.org/10.1186/1472-6963-14-2 PMID: 24383766

- Bailie R, Si D, Shannon C, Semmens J, Rowley K, Scrimgeour DJ, et al. Study protocol: National research partnership to improve primary health care performance and outcomes for Indigenous peoples. BMC Health Serv Res. 2010; 10:129. Epub 2010/05/21. https://doi.org/10.1186/1472-6963-10-129 PMID: 20482810; PubMed Central PMCID: PMC2882384.
- 29. Menzies School of Health Research. Maternal Health Clinical Audit 2014 [cited 2017 24 July]. Available from: https://www.menzies.edu.au/page/Resources/Maternal_health_clinical_audit/.
- **30.** Central Australian Aboriginal Congress. Minymaku Kutju Tjukurpa Women's Business Manual. 4th ed. Alice Springs: Centre for Remote Health; 2008.
- **31.** Three Centres Collaboration. Three Centres Consensus Guidelines on Antenatal Care Project, Mercy Hospital for Women, Southern Health and Women and Children's Health. 2001.
- 32. Bailie RS, Togni SJ, Si D, Robinson G, D'Abbs PHN. Preventive medical care in remote Aboriginal communities in the Northern Territory: A follow-up study of the impact of clinical guidelines, computerised recall and reminder systems, and audit and feedback. BMC Health Serv Res. 2003; 3:1–13.
- **33.** Eades S. Maternal and Child Health Care Services: Actions in the Primary Health Care Setting to Improve the Health of Aboriginal and Torres Strait Islander Women of Childbearing Age, Infants and Young Children. Canberra: 2004.
- **34.** Herceg A. Improving health in Aboriginal and Torres Strait Islander mothers, babies and young children: a literature review. Canberra: Department of Health and Ageing, 2005.
- **35.** National Public Health Partnership. National Aboriginal and Torres Strait Islander Nutrition Strategy and Action Plan 2000–2010 Camberra: Department of Health, 2000.
- **36.** National Aboriginal and Torres Strait Islander child and maternal health exemplar site initiative: site reports 2005. Canberra: Department of Health and Ageing, 2005.
- Cunningham FC, Ferguson-Hill S, Matthews V, Bailie R. Leveraging quality improvement through use of the Systems Assessment Tool in Indigenous primary health care services: a mixed methods study. BMC Health Serv Res. 2016; 16(1):583. https://doi.org/10.1186/s12913-016-1810-y PMID: 27756295
- Bailie J, Laycock A, Matthews V, Bailie R. System-Level Action Required for Wide-Scale Improvement in Quality of Primary Health Care: Synthesis of Feedback from an Interactive Process to Promote Dissemination and Use of Aggregated Quality of Care Data. Frontiers in public health. 2016; 4:86. Epub 2016/05/21. https://doi.org/10.3389/fpubh.2016.00086 PMID: 27200338; PubMed Central PMCID: PMCPmc4854872.
- Huijg JM, Gebhardt WA, Crone MR, Dusseldorp E, Presseau J. Discriminant content validity of a theoretical domains framework questionnaire for use in implementation research. Implement Sci. 2014; 9 (1). https://doi.org/10.1186/1748-5908-9-11 PMID: 24423394
- 40. Schierhout G, Hains J, Si D, Kennedy C, Cox R, Kwedza R, et al. Evaluating the effectiveness of a multi-faceted, multilevel continuous quality improvement program in primary health care: Developing a realist theory of change. Implement Sci. 2013; 8(1). <u>https://doi.org/10.1186/1748-5908-8-119</u> PMID: 24098940
- Wagner EH, Austin BT, Davis C, Hindmarsh M, Schaefer J, Bonomi A. Improving chronic illness care: Translating evidence into action. Health Aff. 2001; 20(6):64–78. https://doi.org/10.1377/hlthaff.20.6.64 PMID: 11816692
- 42. Schierhout G, Matthews V, Connors C, Thompson S, Kwedza R, Kennedy C, et al. Improvement in delivery of type 2 diabetes services differs by mode of care: a retrospective longitudinal analysis in the Aboriginal and Torres Strait Islander Primary Health Care setting. BMC Health Serv Res. 2016; 16 (1):560. https://doi.org/10.1186/s12913-016-1812-9 PMID: 27717351
- Crossland L, Janamian T, Jackson CL. Key elements of high-quality practice organisation in primary health care: A systematic review. Med J Aust. 2014; 201(3):S47–S51. https://doi.org/10.5694/mja14. 00305 PMID: 25047881
- Stewart RJ, Caird J, Oliver K, Oliver S. Patients' and clinicians' research priorities. Health Expect. 2011; 14(4):439–48. https://doi.org/10.1111/j.1369-7625.2010.00648.x PMID: 21176014
- Elwyn G, Crowe S, Fenton M, Firkins L, Versnel J, Walker S, et al. Identifying and prioritizing uncertainties: Patient and clinician engagement in the identification of research questions. J Eval Clin Pract. 2010; 16(3):627–31. https://doi.org/10.1111/j.1365-2753.2009.01262.x PMID: 20482747
- McIntyre S, Novak I, Cusick A. Consensus research priorities for cerebral palsy: A Delphi survey of consumers, researchers, and clinicians. Dev Med Child Neurol. 2010; 52(3):270–5. <u>https://doi.org/10.1111/</u> j.1469-8749.2009.03358.x PMID: 19694780
- Jackson CL, Janamian T, van Weel C, Dunbar JA. Implementation research—Its importance and application in primary care: Moving evidence from theoretical to practical levels requires a collaborative effort. Med J Aust. 2014; 201(3):S42–S3. https://doi.org/10.5694/mja14.00063 PMID: 25047879

- Janamian T, Jackson CL, Dunbar JA. Co-creating value in research: Stakeholders' perspectives: A centre for research excellence reflects on collaborating with policymakers and front-line staff to bring evidence into practice. Med J Aust. 2014; 201(3):S44–S6. https://doi.org/10.5694/mja14.00273 PMID: 25047880
- 49. French SD, Green SE, O'Connor DA, McKenzie JE, Francis JJ, Michie S, et al. Developing theoryinformed behaviour change interventions to implement evidence into practice: a systematic approach using the Theoretical Domains Framework. Implement Sci. 2012; 7(1):38. https://doi.org/10.1186/1748-5908-7-38 PMID: 22531013